



Article Unveiling the Decision-Making Dilemmas in Mini-Grids: The Intricate Case of Smart Meters

Venkata Bandi ^{1,}*¹, Tiia Sahrakorpi ¹, Jukka V. Paatero ² and Risto Lahdelma ^{1,3}¹⁰

- ¹ Department of Mechanical Engineering, School of Engineering, Aalto University, Otakaari 4, 02150 Espoo, Finland
- ² School of Energy System, Lappeenranta-Lahti University of Technology LUT, Skinnarilankatu 34, 53850 Lappeenranta, Finland
- ³ Department of Mathematics and Systems Analysis, School of Science, Aalto University, Otakaari 1, 02150 Espoo, Finland
- * Correspondence: venkata.bandi@aalto.fi

Abstract: Mini-grids need to imitate the transition path of a traditional grid to maintain their position as a sustainable energy access alternative, while aligning with the objectives of the seventh Sustainable Development Goal. One such strategy is implementing smart-metering solutions to improve business viability and remote monitoring of distributed mini-grid assets. However, selecting smart meters presents a significant challenge for mini-grid operators, primarily due to the installation costs involved and the complexities associated with operating mini-grids in rural areas. Against this backdrop, the current case study' demonstrates the utility of multi-criteria decision aids, such as stochastic multi-criteria acceptability analysis (SMAA), to assist mini-grid operators in making informed decisions concerning smart-meter selection. In addition, practitioners' narratives elucidate how implementing smart metering can function as part of mini-grid operations in rural areas. Furthermore, narratives highlight the importance of considering context-specific conditions to avoid the under-utilisation of smart meters.

Keywords: mini-grids; smart meters; MCDA; pre-feasibility study



A recent World Bank report suggests that mini-grids have the potential to provide electricity to 490 million people, as new technologies and falling costs have made them an accessible and viable alternative to a national grid connection [1]. This scenario will require supportive policies, private-sector participation [2,3], a conducive business environment [4,5], and institutional financing [6,7]. In addition, mini-grid operators must adopt smart-grid characteristics to improve financial sustainability and viability [8]. From an energy access perspective, smart-grid characteristics may enable mini-grids to leapfrog the role of traditional power systems in the Global South in the energy access context. Smart grids improve the business viability of mini-grids by facilitating remote monitoring of distributed energy systems [9]. Herein, smart meters play a key role by offering bi-directional communication between the system operator and technical components [10]. Correspondingly, many mini-grid operators have adapted smart meters to control, measure, and manage their assets' technical and economic performance [11]. There is a consensus among energy access practitioners that smart meters could aid in scaling mini-grid business models [12,13]. Despite the importance of smart meters, far too little attention has been paid to their implementation in the existing literature. This article intends to fill this research gap by focusing on smart meter selection and implementation with the help of a case study. The case study setting is a mini-grid company operating in Jharkhand, India.

Selecting the appropriate smart metering setup is crucial for the long-term viability of a mini-grid business, as meters can account for as much as 40% of the total installation costs.



Citation: Bandi, V.; Sahrakorpi, T.; Paatero, J.V.; Lahdelma, R. Unveiling the Decision-Making Dilemmas in Mini-Grids: The Intricate Case of Smart Meters. *Energies* **2023**, *16*, 6112. https://doi.org/10.3390/en16176112

Academic Editors: Mohammad Taufiqul Arif, Md. Abdur Razzak, GM Shafiullah and Kannadasan Raju

Received: 7 June 2023 Revised: 14 August 2023 Accepted: 17 August 2023 Published: 22 August 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). The selection problem resembles a multi-criteria decision problem, as smart meters differ significantly in terms of their characteristics, costs, specifications, and technologies [12]. In such decision-making scenarios, multi-criteria decision aid (MCDA) methods are helpful for decision-makers (DMs) to electrify rural areas through off-grid electrification [13,14]. In practice, MCDA methods support mini-grid operators in choosing from alternatives by considering multiple criteria with conflicting objectives. MCDA tools provide an analytical framework for informed decision-making considering relevant stakeholders' objectives grounded on imprecise information prevalent in real-world problems [15,16]. MCDA has been applied widely to various mini-grid planning and decision problems. Most studies concern energy system planning (e.g., [17–20]). Such pre-feasibility studies are an important tool to evaluate the techno-economic availability of energy resources in the context of community needs, end-user preferences, and the assessment of the social and environmental impacts. However, these pre-feasibility studies seldom account for the practical challenges mini-grid operators encounter in rural areas [21]. These practical challenges include demand uncertainty, limited technical expertise, and techno-economic choices. Moreover, these pre-feasibility studies have not incorporated the selection of smart meters as part of the mini-grid planning. This research gap necessitates a systematic and theoretical analysis of smart meter selection, considering their capital-intensive nature.

Considering the above background, the current article demonstrates the utility of stochastic multi-criteria acceptability analysis (SMAA), an MCDA method [22], to support mini-grid operators for selecting smart meters. The methodological approach involving a single case study aligns with the traditions of MCDA problems in engineering [23]. In addition to presenting the utility of the SMAA approach for smart meter selection, we apply organizational narrative analysis, as proposed by Boje [24], to analyze the interview material. Narrative analysis as a methodological framework facilitates the amalgamation of specific events and processes in a given context [25]. The narratives contextualize practitioners' experiences concerning smart meter use within mini-grid operations in rural areas. Through narrative analysis, interviewees reveal the practical limitations of smart meter installation and maintenance, and the inherent challenges associated with the utility of smart meters in rural settings. Adopting the narrative as a unit of analysis allows us to uncover the nuances and complexities often overlooked in energy research, additionally providing nuances to the available quantitative data [26].

This study offers new insights for mini-grid operators and practitioners. First, the case study catalogues the operational challenges associated with context-specific conditions that may limit the smart-metering approach's utility in rural settings. The reported operational challenges are not adequately addressed in the existing mini-grid literature. Secondly, the study demonstrates the efficacy of MCDA tools in providing valuable insights for the smart meter selection problem, augmenting existing planning studies in the literature. Finally, the hierarchical criteria presented in this study provide a reliable snapshot of the functional aspects of smart metering implementation in rural areas. The proposed hierarchical criteria may be helpful for mini-grid operators as it helps to break down the functional and operational complexities of smart meter implementation into smaller and more understandable characteristics. Overall, this study bridges the existing research gap in the current literature regarding the selection and implementation of smart meters in rural mini-grids.

This article is structured as follows. In Section 2, we provide background on mini-grids and smart meters. Section 3 introduces the case study and data collection, while Section 4 presents the methodology. Section 5 presents the analysis of the qualitative interview material on smart metering implementation. Section 6 presents smart meter selection as an MCDA problem. Before the discussion and conclusion, we illustrate how SMAA is useful for decision making concerning smart meter selection.

2. Smart Meters and Mini-Grid Operations

Mini-grid design and deployment ought to holistically consider economic, technical, and social factors to identify locally appropriate technologies that serve the energy needs of end users in rural settings [27]. From an energy access context, the adopted strategy must align with the local context to navigate the inherent resource constraints associated with designing and implementing mini-grids in rural areas. From a design and implementation standpoint, mini-grids have the following core objectives: electricity generation; energy storage and conversion; end-user consumption; and control, management, and measurement (CMM) [28]. The generation and storage aspects relate to supply reliability. The predominant approach to reliability is identifying viable cost alternatives depending on the local characteristics [13,29]. This approach aims to ensure business viability by balancing the electricity supply and demand. The consumption aspect of the design and implementation strategy focuses on load profile estimation [30] and willingness to pay [31]. The consumption lens pay focuses on ensuring the financial viability of mini-grids. On the other hand, CMM functionality is an essential element of mini-grid planning that addresses numerous challenges, including technological reliability, optimal deployment of constrained resources, demand management, and reduction in supply outages. CMM functionality also addresses other challenges, such as system maintenance, system reliability, business scalability and viability, tariff implementation, and cash flow management. To overcome these challenges, utilities have adopted smart-grid strategies by taking advantage of digital technologies, including smart meters [32,33]. Similarly, Welch et al. supported the smart grid strategy for mini-grids and emphasized the importance of smart meters for addressing operational challenges [34].

Smart meters can provide significant benefits for mini-grid projects in rural areas. They can help address problems with timely rental collection and payment defaults and facilitate the scaling of business operations by offering practical insights into energy use and system reliability [35]. These practical insights encourage DMs to use constrained resources prudently [36]. Smart metering systems also provide tamper protection and risk management [37,38]. Another study argues that implementing smart-metering technologies can mitigate energy theft, reduce social tensions, and improve the viability of off-grid electrification efforts [39]. In addition, smart meters allow different tariff settings for different user groups. These functional characteristics permit shelving unintended new peaks by balancing demand and supply [40]. Taken together, smart meters allow mini-grid operators to address diverse challenges while enabling remote monitoring of day-to-day operations.

The above-mentioned studies present positive aspects of smart metering implementation for mini-grids while considering smart meters as a single technological artefact. This consideration presents an oversimplified view of smart meters as part of mini-grid operations. As a technological artefact, the smart meter is not a single unit; instead, it is part of a configured infrastructure that integrates several technologies, including data communication systems, data management systems, and a web interface to assist with data analytics [41]. Moreover, commercially available smart meters differ significantly in terms of technological configuration. On the other hand, deploying smart meters in rural areas can present additional challenges, such as the lack of reliable network infrastructure for data communication and the limited technical expertise of mini-grid operators. These challenges can be addressed by considering smart meter features and functionalities. The rest of the section briefly overviews smart meter features and functionalities affecting mini-grid operations.

Data communication is the most complex element of a smart-metering setup [42]. Minigrid operators need to test the technological limitations of data communication in a rural setting to reap the maximum utility of smart meters. It is not easy for mini-grid operators to choose the most appropriate technology for reliable data communication, considering the geographical nature of a typical mini-grid. Popular communication technologies include radio-frequency mesh, low data-rate power line communication, and cellular networks. To evaluate these communication technologies, mini-grid planners must consider data latency and outdoor sensitivity when assessing smart meters. Most smart-metering deployments use two layers of data communication. Generally, the higher communication layer facilitates the backhaul of extensive data from the intermediate server to a database. The second layer of the network, commonly known as the data concentrator unit, typically connects the smart meter and the intermediate server. Few deployments use single-layer communication networks to enable communication. Thus, various criteria must be considered while opting for a smart-metering setup to ensure reliable data communication.

Smart meters enable mini-grid operators to account for cash management from electricity distribution. Herein, given the advantages of pay-as-you-go systems, mini-grid operators often prefer to use smart meters in pre-paid mode compared with post-paid operations. In the prepaid mode, a smart-metering setup uses different data communication technologies to transfer credit to the customer's account. In practice, mini-grid operators often rely on local agents to facilitate cash collection owing to poor network reliability concerning data communication in rural areas. In addition, smart meters can help detect and report electricity theft attempts. Electricity theft is a pervasive problem for off-grid system operators (see [43]) and negatively impacts mini-grid business models' sustainability. One study recommends that mini-grid companies safeguard revenues against electricity theft [44]. To this end, mini-grid operators often install smart meters either on distribution poles or outside walls.

Smart meters allow data collection and permit data analytics to gain insights into electricity generation, electricity consumption, customer payments, customer information systems, and overall system performance. Data analytics enable a multi-dimensional analysis of smart-meter data and may provide useful information reflecting the onsite performance of mini-grids and the evaluation of the business model. A recent world bank report suggests that data analytics from smart meters mainly offer two advantages: reduced operational expenditure of mini-grids and improved customer service quality [45]. These two advantages of smart meters could aid mini-grid development prospects concerning universal electrification efforts by 2030.

Smart meters help mini-grid operators improve and monitor their distributed energy assets and business and revenue models, especially remote monitoring functionalities aid in overseeing the operations of dispersed mini-grids suited for rural areas. Herein, the inherent technological sophistication further increases the complexity of decision problems placed on mini-grid operators considering the energy access context owing to resource constraints, as reported in a case study from India [4].

Overall, smart meter selection as a decision problem significantly affects mini-grid operations. In resource-constrained settings such as rural areas, technological choices with high capital costs, such as smart meters, demand an informed decision-making process. In other words, the decision problem—identifying appropriate smart metering setup—concerns multiple criteria, including technological characteristics, remote monitoring capabilities, and operational capabilities of the mini-grid operator. A review of the mini-grid literature elaborates on similar decision problems and multiple criteria for mini-grid operators [46]. For mini-grids, these multi-criteria are characterized by uncertain and imprecise information [18,20]. As a result, DMs can rely on MCDA methods to identify an acceptable solution among different alternatives by considering the relevant criteria [15,47]. In the case of smart meters, different alternatives can be assessed through MCDA methods by evaluating functionalities, characteristics, costs, operational requirements, and specifications in a given setting. An earlier study demonstrates the usefulness of MCDA tools for selecting smart meters in municipal buildings [48].

To summarize, first, smart metering implementation as part of mini-grid operations positively affects universal electrification efforts in rural areas. Second, smart meter implementation requires informed decision making due to high capital costs, technological complexity, and operational advantages. Herein, MCDA tools could aid informed decision making while safeguarding the interests of mini-grid operators in resource-constrained rural settings.

3. Case Study and Data Collection

The case study integrates onsite experiences relating to smart-metering implementation into academic discourse through narrative analysis. It showcases the application of SMAA to assist operators in making informed decisions when selecting a smart-meter setup for mini-grid operations in rural areas. The study setting is a social enterprise operating in Jharkhand, India, whose name has been anonymized by the request of management. The methodology explains the intertwined complexities associated with smart-meter implementation as part of mini-grid operations. Similar case studies are widely used in MCDA studies in engineering and environmental research [23,49]. The rest of the section contextualizes the case study and provides information concerning data collection.

3.1. Case Study Background

Over the last decade, the Indian government has adopted a two-pronged approach to improving electricity access in rural areas. The first approach involves rapidly expanding the national grid to unelectrified villages. The second is the electrification of poor house-holds through the Pradhan Mantri Sahaj Bijli Har Ghar Yojana (Saubhagya) scheme. With this two-pronged approach, the government intended to achieve universal electrification by 31 March 2019. Subsequently, the Indian government declared near-universal household electrification on 31 December 2018 [50]. The International Energy Agency acknowledges rapid electrification while highlighting the poor reliability of electricity in rural areas [51]. In other words, reliability issues persist in rural communities, as end-users experience frequent blackouts. A survey of rural households in 2020 confirmed these reliability issues [52]. Reliability remains a contradiction in the success story of near-universal electrification. The contradiction also prevails in Jharkhand. A survey from 2021 found that rural households in Jharkhand face similar reliability issues with the national grid [53]. Given this scenario, mini-grids may reduce the frequency and duration of power outages, making them more reliable by serving the energy needs of rural communities in the state.

Against the above backdrop, a mini-grid social enterprise began operations in rural Jharkhand in 2017, and it continues to install solar mini-grids to serve underserved communities. Social enterprises—organizations working on unmet needs by balancing profitability and social impact—are suitable for delivering energy services in rural areas considering market imperfections and resource constraints in low-income settings [54]. To assess viability, the enterprise mainly evaluates population size, household energy requirements, consumer willingness to pay, and commercial energy demand, and follows the build, own, operate, and maintain (BOOM) business model. Due to market imperfections, mini-grid operators tend to exhibit a pronounced inclination towards the BOOM business model, making it a natural choice [55]. Unlike the national grid, where multiple organizations are responsible for decision-making, the BOOM business model requires mini-grid operators to oversee a myriad of decisions concerning technological choices, operations management, resource allocation, revenue collection, and customer management. The complexity concerning the decision-making process presents paradoxical challenges surrounding how different sociotechnical aspects of the mini-grid are designed, managed, and operated [4].

By the end of 2018, the company had installed 40 mini-grids and was planning to install 100 additional mini-grids by 2023. Statistically, the company intended to install four mini-grids every quarter. As with many social enterprises, the company inherently faced resource constraints. These constraints often arise due to various factors, including a necessity to focus on social impact, limited access to financing, and in-house technical expertise. To support the company personnel, a team of experts, which included the first author, provided technical assistance to the enterprise in various areas, such as smart meters, energy-efficient devices, storage options, and process improvements from 2018 to 2019. The technical assistance was financed by the Good Energies Foundation as part of a capacity-building project. The focus of the current article is limited to smart meters.

Since the beginning of solar mini-grid installations in Jharkhand, the enterprise has incorporated smart meters into its technical setup. However, the initial choice of smart meter technology imported from Spain created challenges during onsite deployment, prompting the company to switch to an alternative smart-metering setup. At the time of the capacity building project, the company was facing operational challenges, which were preventing them from realizing the full potential of its smart metering setup. To this end, the management team sought assistance from an expert team to determine the most suitable smart-metering system for their future mini-grids. Such collaborative partnerships are essential in resource-constrained entrepreneurial settings [56,57].

3.2. Data Collection

From the capacity-building project standpoint, the main objective of decision problems was to identify the most suitable smart meter(s) based on the preferences of the mini-grid operator. The expert team assisted the management team in evaluating appropriate smart-metering options for solar mini-grids. The assistance involved proving technical expertise for comparing various alternatives, with the current metering setup as a reference.

To unpack the complexity of decision problem, the expert team gathered qualitative and quantitative information on the company's use of smart meters using a semi-structured questionnaire. The questionnaire was tailored to the respondents' experiences with smart meter implementation, allowing for open-ended responses. In practice, during the semistructured interviews, all participants were asked the same set of questions, and specific additional questions were included based on the interviewee's occupation. The standard questions were as follows:

- What installation and operational challenges are typically associated with smart meter implementation?
- What are the main technical challenges of smart meter implementation, and how are these challenges affecting mini-grid operations?
- Are you satisfied with the existing process concerning prepaid transactions?
- What smart meter functionalities are essential for mini-grid operations?
- Taking into consideration the overall costs involved, how would you evaluate the performance of the current implementation of smart meters?
- In your opinion, what specific modifications or changes are needed in the current smart meter implementation to address the existing operational challenges effectively?
- What are the main concerns reported by ground-level technicians concerning smart meters?

The data-gathering process lasted four months (December 2018 to March 2019) and included stakeholder interviews, data collection from site visits, identification of alternatives, and the presentation of results to DMs. Stakeholders involved in the data collection process included plant engineers, field technicians, the technical team overseeing metering implementation, the management team (as DMs), and other experts. Upon completing the data collection process, the technical and management teams identified multiple criteria in collaboration with the expert team at the company headquarters in Kolkata. In April 2019, the expert team suggested nine alternatives to the existing metering setup, and the management team selected four alternatives for further evaluation. The management team discarded two alternatives were rejected mainly due to capital cost. The expert team then presented a comparative analysis of the remaining alternatives, including the existing setup, in June 2019. The comparative analysis provided an overview of each alternative without MCDA support.

In this article, the authors expand on the original decision problem by utilizing the data collected during the project period to showcase the application of SMAA to ranking alternatives based on multiple criteria analysis to aid DMs. In addition, the current article presents an analysis of the qualitative data using narrative analysis. The narrative

analysis highlights invisible nuances by elucidating the overarching themes [58]. The interviews at company headquarters, specifically with DMs, focused on organizational capabilities and overall utility in the purview of smart metering implementation. All interviews were recorded, transcribed, and analyzed. Table 1 presents information related to the respondents.

Table 1. Qualitative interviews.

Title	Identifier	Language(s) of Narrative
Regional manager	А	English and Hindi
Assistant manager	В	English and Hindi
Technical manager	С	English and Hindi
Assistant technical manager	D	English and Hindi
Deputy country director	E	English

4. Methods and Methodology

This case study used a multi-disciplinary approach, a combination of SMAA and narrative analysis, to achieve its research objectives. The current section explains these two methods.

4.1. Stochastic Multi-Criteria Acceptability Analysis (SMAA)

Concerning the social enterprise's decision problem, piloting is one way to compare different smart meters. The piloting process involves the implementation of new technology for testing and evaluation purposes. However, piloting is expensive for mini-grid companies in rural, resource-constrained settings. Another approach is to rely upon descriptive methods without any methodology. This approach is inadequate for structuring complex problems involving conflicting objectives. Therefore, a more suitable option is to compare different meters using MCDA methods.

MCDA methods are useful in decision-making processes to manage imprecise information, criteria, uncertainties, and DM preferences, thereby replacing the intuitive selection of alternatives with justified decision models. To analyse the decision problem in this instance, we opted for SMAA [22]. SMAA is an MCDA tool for evaluating alternatives regarding multiple non-commensurate criteria [59]. SMAA was selected as it can be applied to decision problems with ordinal and cardinal criteria, where information is imprecise or uncertain. This section briefly describes the SMAA method. For a detailed overview and application of SMAA, see [60].

SMAA relies on Monte Carlo simulations to produce MCDA parameters. The simulation model randomly generates criteria measurements and weights from their corresponding distributions at each simulation round. One study recommends the number of iterations to be between 10,000 and 100,000 [61]. The current study involves 10,000 iterations. The alternatives are ranked based on utility function values, and the corresponding evaluation statistics are collected. Preference information can be considered flexibly in SMAA. Monte Carlo simulation rejects weights that do not satisfy the constraints defined in the MCDA model.

SMAA methodology helps DMs choose their most preferred alternative from a discrete set of *r* alternatives $x = \{x_1, x_2, ..., x_r\}$. These alternatives are evaluated based on multiple criteria, where *s* is the number of criteria. The decision model combines the criteria measurements with DMs' preferences by calculating a utility function $u(x_r, w)$, where *w* is a set of subjective weights given by DMs concerning different criteria. In simple terms, the utility function expresses different criteria with a numerical value to represent the preference for each combination of criteria. The utility function can have different shapes. Current SMAA implementation uses an additive function with partial utility in this case. For a detailed description of additive partial utility, refer to [62]. The additive

utility function computes the overall utility as a weighted average of the partial utilities of alternatives:

$$u_r = u(x_r, w) = \sum_s w_s u_s(x_{rs})$$

The partial utility function $u_r(x_{rs})$ maps the criteria measurements to the interval [0, 1], where 0 is the worst value and 1 is the best value. The weights are normalised in SMAA to be non-negative, and their sum is 1. The set of feasible weights is defined below:

$$W = \{ \boldsymbol{w} | w_s \ge 0 \text{ and } \sum_s w_s = 1 \}$$

As mentioned in the introduction, SMAA is a useful MCDA tool for aiding DMs in dealing with uncertain criteria measurements and approximate weights. Stochastic variables represent uncertain criteria measurements $[x_{rs}]$ using the joint density function $f_x(x)$. Similarly, the joint density function $f_w(w)$ represents imprecisely known weights. If weight information is missing, then SMAA uses uniform distribution.

SMAA, as a method, accommodates ordinal and cardinal criteria measurements [63]. This versatility offers strategic flexibility in the decision-making process. However, measuring cardinal values for each criterion is often expensive and time consuming. Despite ambitious efforts, the measurements remain uncertain or inaccurate for many real-life decision problems. Ordinal criteria measurements are helpful in this respect, as they allow a combination of qualitative and quantitative information to be represented by ranking the alternatives. SMAA treats ordinal criteria in the following manner. For example, suppose alternative A is better than C, which is better than B concerning some criterion. In that case, SMAA assigns a partial utility of 1 for alternative A and 0 for alternative B, and randomly generates a utility value in the interval [0, 1] for alternative C following a uniform distribution. See [61] for more details.

MCDA methods often use a criteria hierarchy to deal with the superordinate and subordinate relations between different criteria. Similar criteria often lead to sub-criteria at successively lower levels, forming a hierarchical tree structure. In this application, we extended the SMAA method to treat hierarchical criteria. The root node of the criteria hierarchy was j = 0. The main-level criteria were J(0). Leaf nodes in the hierarchy represented the lowest-level criteria without sub-criteria. The partial utility of the lowest-level criteria without sub-criteria. The partial utility of the lowest-level criteria the measurement to the range [0, 1], where 0 was the worst and 1 was the best value:

$$u_r^j = \begin{cases} \sum\limits_{k \in J(j)} w_k u_r^k & \text{When node } j \text{ had sub} - \text{criteria } J(j) \\ u_j(x_{r,j}) & \text{and node } j \text{ was a leaf node} \end{cases}$$

In the case of a criteria hierarchy, the weights (w_j) were normalised separately at each non-leaf node:

$$W^j = \{ oldsymbol{w} | w_k \geq 0 \ and \sum_{k \in J(j)} w_k = 1 \}$$

SMAA calculates various descriptive measures for assessing the alternatives. The main measures are explained below:

- The acceptability index (a_r) measures the share of different valuations making the alternative x_r a preferred alternative. Acceptability indices categorise alternatives stochastically within the range [0, 1]. A near-zero acceptability index (e.g., $a_r < 0.05$) is never considered as best with the assumed MCDA model.
- The rank acceptability index (b_r^i) generalises the acceptability index by measuring various weights that grant alternative x_r rank *i*. A near-zero rank acceptability index indicates that the alternative will never obtain rank *i*. An alternative with high acceptability for the worst ranks should be avoided.

- The central weight vector (w_r^c) characterises the expected centre of gravity of an alternative x_r when it is most preferred. The central weights without preferential criteria allow for an inverse approach to MCDA. In this case, the central weights allowed DMs to understand how different weights corresponded to different choices and made them preferred alternatives [64]. In general, the central weights are undefined for incompetent alternatives.
- The pair-wise winning index (*c_{ij}*) is the probability for alternative *i* to be more preferred than alternative *j*. The pair-wise winning index helps exclude alternatives that are dominated by others.

4.2. Validity of SMAA Results

To ensure the validity of MCDA results, the chosen method ought to align appropriately with the decision problem. Moreover, the problem must be well-defined in terms of both criteria and alternatives. Lastly, accurate measurements of criteria and preferences are crucial factors for maintaining the validity of the analysis. However, MCDA models often need to deal with incomplete, imprecise, and uncertain information on the evaluations and preference model parameters.

SMAA was selected for the smart meter selection problem, because of its flexibility in handling mixed ordinal and imprecise cardinal criteria, and uncertain or absent preference information. Moreover, the validity of SMAA, as a MCDA method, is supported by existing real-life applications in academic literature (see examples: choice of district heating [65], renewable power production at Moroccan airports [66]). A recent acdemic work reviewing SMAA methods of applications presents the theoretical development of SMAA and its robustness by evaluating 118 published articles in scientific lierature [60].

In addition, a team of experts collaborated with DMs to establish the alternatives and criteria. Likewise, the set of criteria used in the SMAA model comprehensively captured all the features relevant for mini-grid operations. In addition, the criteria met the general requirements for model validity specified by Keeney and Raiffa [67] for decision problems with multiple objectives and different trade-offs: completeness, operationality, non-redundancy, and minimality.

4.3. Narrative Analysis

Numerous qualitative data analysis methods, such as grounded theory, thematic analysis, and discourse analysis, are available to analyse semi-structured interviews. In this instance, we opted for narrative analysis owing to its ability to capture otherwise invisible nuances within organisational settings [68]. Moreover, semi-structured interview responses usually take a narrative format, where respondents recollect events in sequence and with consequences. In narrative analysis, the recollected events were 'selected, organised, connected and evaluated' [69]. The narratives illuminate how different DMs in the rural social enterprise reflect on the smart-metering setup. The main limitation of narrative analysis is that the interview setting influences narratives and to whom the narrator is talking [70] (p. 180).

As a research method, narrative analysis provides critical insights for energy research [26,71] and has become increasingly popular among social science researchers. As mentioned in the introduction, for this study, we relied on Boje's approach to narrative analysis, which considers organisations as storytelling systems [24]. Practically, people within an organisation recollect organisational endeavours and recount stories of success and failure concerning different events and processes. Collected data was organised to identify meaningful patterns for elucidating how the smart-metering setup adopted functions as part of mini-grid operations at a grass-roots level.

Essentially, the narrative analysis employed in this article considers the operational experiences of the mini-grid enterprise as events with inherent sequences and consequences. The experiences on the field level, where unexpected and unwarranted operational complexities unfold, accentuate invisible tension points in mini-grid operations. Often, mini-grid literature seldom features these rural onsite experiences, as grander narratives predominantly focus on positive narratives of smart meter diffusion in rural settings on a macro level. Against this background, the narratives presented in this article may help minigrid practitioners adopt strategies for achieving smart-meter implementation objectives in rural settings.

To summarize, the current study relies on SMAA to evaluate smart metering alternatives and employs narrative analysis as the qualitative data analysis method to contextualize the onsite experiences of the mini-grid operator. SMAA was selected for MCDA analysis because it can handle imprecise information concerning multi-variate decision problems, whereas narrative analysis was chosen because it offers greater depth, explanation and meaning for analysing qualitative material.

5. Smart Meters: A Decision-Making Dilemma

Mini-grid operators have benefited from significant improvements in the quality of system components, such as photovoltaic panels, batteries, and inverters, over the last decade [45]. Although smart meters have become more widespread in recent years, their effectiveness varies depending on their overall utility in rural settings, as indicated in the narrative analysis presented below. As the following accounts suggest, the company faces various operational challenges despite substantial investments in its smart-metering setup. The implementation of smart-metering architecture has failed to deliver the promised advantages for the company's mini-grid operations. Our narrative analysis reveals that the metering architecture was misaligned with the heterogeneity of mini-grid operations in rural areas at the time of the study.

For two reasons, pre-paid energy meters are essential for the company's mini-grid operations. First, the pre-paid business model avoids social tension in rural settings. Second, this is the easiest payment model option for low-income rural consumers. When asked about the objective of smart meters, Respondent A stated, "Smart metres with pre-paid system reduces conflicts with customers on bill payments, especially here [referring to rural areas]", and continued, "With smart metres, we can compare cash coming from different sites with the [smart meter web portal]". By framing their response in this manner, Respondent A stressed the importance of prepaid smart meters as the most efficient way to monitor revenues from cash transactions. Respondent B expressed a similar view on prepaid meters, "Our business model works only with prepaid mode; otherwise, bill collection will be difficult in rural areas". In other words, the company opted for smart meters to obtain the benefits of prepaid functionality. In addition, the regional manager stressed the importance of revenue monitoring features within a smart metering setup. These narratives aligned with conventional prepaid mini-grid business models within a rural setting.

The company has employed a smart-metering system that utilizes a 20-digit vend code and a mobile application based on Bluetooth technology to enable prepaid transactions. The company must register plant engineers' mobile numbers with the smart metering web portal to implement automatic prepaid energy credits transfer to end users. The registered mobile numbers will receive a 20-digit vend code via the mobile application, which can be prepaid credits to a specific smart meter corresponding to unique customers. The smart metering setup uses a customer information system for identifying energy meters corresponding to unique end users. The mobile application ensures secure digital transactions and minimizes fraudulent activities. However, the digital transaction will only be completed when the registered mobile phone is near the smart meter, within a range of 10 m. Then, the mobile application uses Bluetooth technology to establish a secure connection with the smart meter and enables the transfer of prepaid energy credits to end users.

The following narratives suggest that the company has not effectively utilised the full potential of the smart metering setup described above. Respondent C explained the downside of the existing top-up process for prepaid transactions, "Our plant engineers,

with the help of local staff, manually enter the 20-digit vend code to transfer pre-paidenergy credits. They carry a ladder to input the code as meters are mounted on the pole" (Figure 1). The respondent added that, in addition, "the mobile application does not work as we imagined". This narrative illustrates the drudgery involved in recharging, which seems to have surprised the company personnel in contrast with the pre-paid process described above. Respondent D, who works with plant engineers, explained the rationale behind the drudgery, as follows:



Figure 1. Manual top-up process.

"The mobile application [refers to Bluetooth application] is registered with a mobile number [of a plant engineer]. We need to contact [the smart meter manufacturer] every time the plant engineer number changes. This is time taking, so we avoid contacting [the manufacturer]. Our plant engineers need to go near the pole for the mobile application to work in any case. So they don't complain about the manual entry [of 20-digit vend code] as local staff helps them".

The above narrative provides a positive interpretation of operational challenges posed by the smart-metering setup, indicating that the perception of these issues varies depending on the role of the employee in question. Moreover, the assistant managers' account indicates the limitations of mobile applications from a practical perspective, as Bluetooth works only within a short range, up to 10 m. This distance limitation creates another operational challenge for the plant engineer, who must know the customer's location in advance. When asked about the significance of the plant engineer's mobile number concerning the ineffectiveness of the Bluetooth mobile application, Respondent D stated:

"Initially, our company gave mobile phones [with pre-paid connections] to plant engineers with the Bluetooth application installed. This proved problematic when people resigned from the company [which required the mobile phone to be collected and handed over to a new plant engineer]. So, we asked them [referring to new plant engineers] to install the application on their mobile phones. Most of them do not want to install the application as it affects mobile performance. I have experienced it personally as our [mini-grid] sites have more than 100 customers".

Initially, when the company had few plants, it was not difficult for management to ensure that the mobile phones were handed over to new plant engineers. However, as operations began to expand rapidly, management instead requested the plant engineers install the Bluetooth application on personal mobile devices to avoid the inventory management of mobile phones. Based on the narrative, plant engineers are hesitant to use the mobile application as it can negatively impact their mobile device's performance owing to having many customers. The negative impact, in turn, added an additional layer of complexity concerning smart meter use at the plant level, stemming from unforeseen complexities of technical configuration. Echoing this plant-level operational challenge, Respondent C highlighted an operational challenge with the existing pre-paid process:

"At present, the top-up [or pre-paid] process requires manual intervention despite having smart meters. The plant engineer sends a [WhatsApp] message to the regional office for top-up, and an operator [with login credentials] will top-up using the customer meter number. The generated [vend] code is sent back to the plant engineer [via WhatsApp message] for a top-up".

The above narrative implies that the company relies on a manual top-up procedure, despite investing in smart meters. As a result of the above-mentioned operational challenges, WhatsApp is utilised for communications related to pre-paid requests and vend codes. All interviewees implied that the manual process resulted in mistakes, creating further operational challenges.

Based on the following narratives, weather conditions in Jharkhand pose another challenge to the company. Respondent E commented: "Some of our metres often get damaged due to frequent lightning in this area. We sought help from our supplier too. Their engineers could not offer any solution". This issue was also mentioned by Respondent C, who clarified that it is not only smart meters that are affected by lightning, but a variety of plant components. The supplier of smart meters suggested improving lightning protection for poles with smart meters (See Figure 1). However, this solution is not necessarily feasible due to the cost and dispersion of rural end users.

6. Decoding Smart Meter Characteristics Using Multi-Criteria

The narrative analysis presented above suggests that the existing smart-metering setup created unforeseen operational challenges for mini-grid operators. These challenges could further complicate mini-grid operations, which are inherently complex. As the narratives reveal, the mini-grid company underutilises smart meters as the operators in rural markets have limited knowledge of the diverse characteristics and functionalities of smart-metering solutions. In such situations, a smart-metering setup should preferably be included in the pre-feasibility analysis. Here, mini-grid operators could use MCDA tools as part of pre-feasibility studies to aid in evaluating different smart-metering options.

As mentioned in Section 4, DMs selected these five alternatives including the existing smart metering setup (SM1). Five smart meter alternatives are listed in Table 2 with their abbreviations and descriptions. While SM1 and SM2 are immediately available, other meters require advanced purchase orders to avoid logistic delays.

Table 2. Smart meter alternatives.

Abbreviation	Communication Technology for Top-Up	Location	Availability
SM1	Bluetooth and mobile application	Pole	Mass-produced
SM2	Cellular and mobile application	Pole	Mass-produced
SM3	RF and cellular with mobile application	Pole	Purchase order
SM4	RF and cellular with intermediate server	Pole	Purchase order
SM5	Field-area network-enabled smart meter with intermediate server	Ground mount	Purchase order

During the capacity building project, the expert team evaluated the alternatives without using MCDA tools, as mentioned earlier. When considering the selection of smart meters, it is important to take into account four intuitive criteria for MCDA: economic factors, the top-up process, metering setup, and functionality. However, the narratives surrounding operational challenges underscore the significance of two additional non-intuitive criteria: usability and design. These two criteria encompass the overall effectiveness of smart meters. From a design perspective, smart metering implementation in rural areas

smart meters. From a design perspective, smart metering implementation in rural areas often requires customized solutions with greater flexibility due to dispersed households and unreliable communication technologies. Customization and flexibility can ensure that the smart metering system is optimized for the context-specific requirements of rural areas. The usability of a smart metering setup affects its performance and justifies its capital costs. To this end, the usability criteria should also consider the suitability for the local context, service offered by the supplier, overall complexity, technological maturity, and expertise required.

Table 3 lists six main criteria considered for informed decision making. The main criteria are further broken down into 19 sub-criteria within a hierarchical structure to capture the functional, operational, and technical characteristics of smart meters in rural settings. While the economic criterion was measured on a cardinal scale, the other criteria were assessed only on an ordinal scale. The utilization of ordinal scales is widely prevalent in MCDA problems. To rank the alternatives, the project team relied on their expert judgment and made ordinal assessments. The criteria measurements for SMAA models are presented in Table 4, with the sub-criteria being measured ordinally, except for the economic criteria. Ordinal criteria focus on the qualitative or subjective judgment of DMs and experts. In this article, ordinal criteria were evaluated considering the overall variance perceived by the authors. In addition, this study drew upon earlier research conducted on smart meters as a reference for ordinal criteria [47]. For example, the overall complexity of a smart metering setup was measured with a scale ranging between 1 and 4, whereas flexibility from a design perspective was measured with the scale ranging between 1 and 5.

Main Criteria	Sub-Criteria	Explanation		
Economic	Е	Annuity on initial investment and total operational costs		
Тор-ир	T1	Human intervention required for top-up		
	T2	Ease of accounting top-ups across multiple mini-grids		
	T3	Time required for handling top-ups		
	T4	Outdoor sensitivity of communication technology		
	M1	Ease of repair and replacement in the case of meter faults		
Metering setup	M2	Ease of installation		
	M3	Modification required for metering enclosures compared to the reference scenario		
Functionality	F1	Monthly reports for evaluating individual plant performance		
	F2	Customized reports with infographics combining the performance of multiple plants		
	F3	The effort required to modify tariff and load limits		
	F4	Periodic alerts to aid mini-grid operators		
Usability	U1	Suitability for the local landscape and built environment		
	U2	Service request handling capacity of the supplier		
	U3	Overall system complexity		
	U4	Technological maturity of the system		
	U5	Operational expertise required for installation and maintenance		
Design	D1	Level of customization for mini-grid operations and rural settings		
Design	D2	Flexibility from an organizational perspective		

Table 3. Main criteria and sub-criteria for smart meter evaluation.

Main Criteria	Sub-Criteria Weights	SM1	SM2	SM3	SM4	SM5
Economic	1.00 (E)	$500\pm10\%$	$580\pm10\%$	$430\pm10\%$	$470\pm10\%$	$448.5\pm10\%$
Тор-ир	0.40 (T1)	5	3	1	1	2
	0.40 (T2)	5	5	1	3	3
	0.10 (T3)	5	1	1	3	3
	0.10 (T4)	1	3	3	2	5
Metering setup	0.50 (M1)	1	1	4	2	1
	0.25 (M2)	1	1	4	2	3
	0.25 (M3)	1	1	4	2	3
Functionality	0.40 (F1)	3	1	2	1	1
	0.10 (F2)	3	2	2	1	1
Functionality	0.40 (F3)	3	2	2	1	1
	0.10 (F4)	3	3	1	2	1
Usability	0.25 (U1)	1	3	2	3	4
	0.20 (U2)	1	1	3	2	4
	0.15 (U3)	1	2	3	2	4
	0.15 (U4)	1	1	4	3	1
	0.25 (U5)	1	2	4	2	3
Design	0.40 (D1)	5	3	2	2	1
	0.60 (D2)	1	1	5	3	2

Table 4. Criteria measurements with weights given by operational team and the preference information of the management team.

Preference information for main criteria as per the management team: Economy > Top-up, Metering setup > Functionality, Usability > Design.

Economic criteria are an essential consideration when making decisions in resourceconstrained settings. In a hierarchical model, economic criteria encompass capital costs and operational costs. The SMAA model presents the economic criterion using the annuity factor by combining investment costs and total operational costs for five years. The investment cost (I) for each smart-metering setup was estimated based on price information provided by the supplier, which includes the cost of the metering enclosures and protection circuitry. The operational costs (O) were computed considering communication topology and information provided by the supplier. For the total cost (C), if financed by an effective interest rate of r for n years, then the annuity of total economic costs A is calculated as follows:

$$A = \left(\frac{1}{\sum_{i=1}^{n} (1+r)^{i}}\right)C, \text{ where } C = I + O$$

The prepaid mechanism offers end-users flexibility and allows customers to purchase electricity in small amounts. As a functionality, this top-up process plays an instrumental role in mini-grid operations. Thus, including the top-up process as part of the MCDA analysis is important. To this end, based on the narrative analysis and the company personnel inputs, we identified four sub-criteria—required human intervention, ease of top-up, time required for a top-up, and outdoor sensitivity of communication technology. Given the diverse nature of the top-up process, these criteria are represented ordinally between 1 to 5.

The metering setup criterion considers installation and maintenance aspects of smart meters as a socio-technical configuration in rural areas. In practical terms, metering setup, as the main criterion, focused on ease of installation, repair, replacement, and modification of existing enclosures (see Figure 1). Alternatives were ordinally measured against the existing metering setup. These criteria were ranked between 1 to 4.

Smart-metering solutions, by design, can perform a range of functional tasks, where tasks are designed to accomplish particular undertakings from a sociotechnical design perspective. Based on the available qualitative material, the mini-grid company identified

four essential functions and prepaid functionality. They are monthly reports at the plant level, customized reports that combine the performance of different plants, the ability to modify tariffs and loads to cater to changing business needs, and periodic alerts reflecting the company's overall performance. Unlike the economic criterion, these criteria were measured ordinally.

The usability of smart meters is a crucial factor affecting mini-grid operations in rural areas. In the SMAA model, usability is defined as the functional aspects of a smart meter within a specific setting, considering the limitations faced by mini-grid operators. Usability is often included in MCDA problems, and this case study identifies several sub-criteria that contribute to the usability of smart meters, including system complexity, technological maturity, and operational expertise. Due to usability's subjective and immeasurable nature, sub-criteria are often measured ordinally.

As the narrative analysis revealed, rural mini-grid settings need a distinct technology design perspective to address issues such as inadequate worker skills and resource constraints. Thus, we considered design as an additional criterion to measure the suitability of smart-metering solutions for the rural mini-grid market segment. In practice, this hierarchical criterion assessed the level of customisation offered by each smart-metering alternative and overall flexibility from an organizational perspective. Given the diverse nature of different smart meters, these criteria were ranked between 1 to 5.

7. SMAA Analysis: Evaluating Smart Metering Alternatives

The evaluation of various smart metering alternatives involved the application of the MCDA method in three phases. Each phase of SMAA evaluation involved 10,000 iterations to ensure a robust analysis. This phase-wise approach with 10,000 iterations ensured a more thorough and reliable assessment of the alternatives [62]. The first phase involved performing an SMAA analysis using the analytical hierarchy methodology with uniformly distributed central weights without considering the decision-makers preferences. Here, the uniform distribution represented incomplete preference information concerning real-world problems [68]. In the second phase, the MCDA model considered normalized weights for sub-criteria (see Table 4) based on the inputs provided by the operations team. The main criteria weights were unconstrained in this analysis. The third-phase model used preference information (see Table 4) provided by the DMs as the main criteria. As per management, economic criteria outweighed all of the other criteria, whereas design aspects warranted the least preference.

7.1. MCDA Model without Preference Information

Figure 2 presents the rank acceptability indices (b1, b2, b3, b4, and b5) for the smartmetering alternatives based on the analytical hierarchy methodology without preference information. The figure shows that SM4 received the highest acceptability (37.1%) for the first rank (b1), indicating that it may have been the preferred choice. SM1 received a rank acceptability of 30.8% for the first rank. The proximity between the rank acceptability of SM1 and SM4 indicated the need for subsequent analysis, as the preference information of the DMs had the potential to alter ranks. SM2 received a first-rank acceptability of 15.9%, followed by SM5 (14%) and SM3 (2.2%). Notably, SM2 received similar rank acceptability indices for the second, third, and fourth ranks. The similarity may have been due to SM2 being functionally better than all of the other alternatives. The dominance of functionality is apparent from the central weight in the case of SM2 (see Figure 3). The central weights revealed what kinds of preferences were favorable for each alternative. For example, alternative SM4 was favored by a high weight on top-up procedures (51.6%), while SM5 was favored by an emphasis on the top-up procedure (28.2%) and functionality (23.8%). In addition, the inconclusiveness of the model is apparent from the pair-wise winning indices (see Table 5). For example, SM1 was a better alternative than the other smart meters, with probabilities of 48.22%, 73.95%, 48.36%, and 43.92%. In other words, this MCDA model indicated the need for subsequent analysis, as no alternative predominated each other. The



need for subsequent analysis is apparent from the pair-wise winning indices of SM1, SM2, and SM4 in Table 5.

Figure 2. Rank acceptability indices (b1 to b5) of alternatives expressed as percentages without incorporating decision makers' preferences.



Figure 3. Central weights (expressed as %) for the SMAA model without preference information.

Alt	SM1	SM2	SM3	SM4	SM5
SM1	0	97.16 (48.22)	96.14 (73.95)	56.36 (43.92)	83.47 (48.36)
SM2	2.84 (51.78)	0	60.72 (79.06)	4.55 (41.21)	24.1 (43.47)
SM3	3.86 (26.05)	39.28 (20.94)	0	1.69 (8.41)	15.34 (5.59)
SM4	43.64 (56.08)	95.45 (58.79)	98.31 (98.59)	0	83.17 (58.69)
SM5	16.53 (51.64)	75.9 (56.53)	84.66 (94.41)	16.83 (41.31)	0

Table 5. Pair-wise winning indices.

After the first phase of analysis, the main objective was to consider how different subcriteria affected the decision problem, particularly how the information provided by the operations team affected smart-meter selection. The qualitative information was quantified as preference information in the case of different sub-criteria with the support of an expert team. Figure 4 presents the rank acceptability indices from the operations team perspective. The first-rank acceptability of SM4 improved to some degree ($37.4\% \rightarrow 43.1\%$), while SM1, SM2, and SM5 were still relevant, as their first-rank acceptability remained significant from an MCDA perspective. In other words, these three alternatives were competitive with SM4. With the preferences of the operations team taken into account, the result for SM3 was very similar to that obtained from the first model. Overall, SM4 was the strongest candidate for the first rank in the first two models, followed by SM1. If the mini-grid company continues its existing setup, then SM4 will be eliminated. In addition, the pair-wise winning indices from Model 2 were similar to those from Model 1. As a result, we have not presented the pair-wise indices from Model 2 in this article. Overall, this model also suggests the need for subsequent analysis.



Figure 4. Rank acceptability indices (b1 to b5) of alternatives expressed as percentages with the preference information of operational team for sub-criteria.

7.3. MCDA Model with Preference Information of the Management Team for Main Criteria

The third SMAA model considered the preference information provided by the management team as DMs for the main criteria. The preference information was represented ordinally in the model (see Table 4). The management team gave the highest importance to the economic criterion, followed by the top-up procedure, while the design aspects were given the least importance. The other two criteria received equal importance. Figure 5 presents the rank acceptability indices for each alternative. Compared to the previous models, the rank acceptability indices for this model indicate a clear pattern. For example, SM2 and SM3 received 0% acceptability for the first rank, and both are unfeasible alternatives. The first-rank acceptability of SM4 improved significantly due to the emphasis on the economic criterion. Moreover, SM1 and SM4 clearly dominated over all the other alternatives. The dominance is apparent from the pair-wise winning indices in the case of SM1 and SM4 (Table 5), which show that these metering setups were considered superior to all other alternatives. For example, the probability of SM1 being preferred over SM2 was 97.2%. Similar observations can be observed in the case of SM4.



Figure 5. Rank acceptability indices (b1 to b5) of alternatives expressed as percentages with the preference information of the management team for the main criteria.

8. Discussion

Much of the conversation around mini-grids highlights smart meters' importance and advantages in rural settings. There is little discussion in the literature about how the choice of smart meters functions at a plant level in resource-constrained settings. Against this background, this article used a case study to showcase the application of SMAA, an MCDA tool, to aid DMs in evaluating smart-metering solutions. The SMAA method was applied to rank different smart metering alternatives. The MCDA approach involved a set of subcriteria, including, but not confined to, economic criteria. In addition, we drew on onsite experiences to illuminate the limitations of smart metering in low-income rural settings utilizing narrative analysis. This multi-disciplinary approach—combining qualitative and quantitative research methods—provided the following insights.

In general, widespread optimism and consensus prevail in mini-grid literature concerning the role of smart metering infrastructure as a key component for business viability. Despite this broad agreement, there are still gaps between the potential for smart meters in low-income settings and their implementation in rural areas, as suggested by the narrative analysis, which has revealed the following from a sociotechnical perspective.

First, the narratives illuminate the unforeseen operational challenges the mini-grid social enterprise faces. A myriad of sociotechnical factors have contributed to these operational challenges. For example, plant engineers are reluctant to use the Bluetooth mobile application, citing a negative effect on their mobile phone performance. As a result, the company personnel continue to rely on manual top-up processes. The reliance on manual operations and infrequent mobile application usage limits the remote monitoring capabilities and necessitates manual accounting of revenues. Taken together, narratives dispute the notion that smart metering aids in addressing operational challenges in the rural mini-grid environment. To avoid these unforeseen challenges, mini-grid operators need to consider the context-specific conditions that will influence their smart-metering setup's utility in order to minimize unintended consequences.

Second, the narratives indicate that mini-grid operators under-utilize the benefits of a smart-metering setup. Under-utilization implies inefficient use of scarce financial resources, as smart meters account for up to 40% of capital costs. Herein, inefficiency threatens the viability of mini-grid in low-income rural markets, as mini-grid operators often face considerable financial dilemmas when allocating scarce resources. Furthermore, although smart meters are considered crucial for the shift towards smart mini-grids, it is important

to note that the successful implementation of smart metering requires thorough analysis and consideration during the planning phase. As the capacities of social enterprises are generally limited, the planning phase will aid mini-grid operators in evaluating smart metering alternatives. To this end, mini-grid operators could benefit from the criteria identified in the case study. The criteria present several functional aspects of a smart metering installation in rural settings. This functional view, presented as hierarchical criteria (see Table 3), counters the over-simplified representation of smart meters in the current mini-grid discourse.

In addition to the narrative analysis, the study establishes the usefulness of the MCDA method for identifying smart meters for mini-grid operations in rural areas. Concerning the social enterprise's decision problem to examine context-specific conditions, the MCDA approach has advantages over descriptive methods and field piloting. This case study demonstrates that selecting a smart-metering setup is a multi-criteria decision problem with varied and conflicting objectives. Given this decision problem, our approach illustrates the advantage of utilizing SMAA to evaluate different criteria associated with smart-metering installation in rural areas. Specifically, SMAA problem formulation emphasizes the importance of a hierarchy of different criteria to capture the dissimilarity of smart meters in terms of the characteristics, costs, specifications, and technologies.

9. Conclusions

The current article sets out to demonstrate the utility of SMAA to make informed decisions while selecting smart meters for mini-grids in rural areas. In addition to presenting the utility of the SMAA, the academic endeavor attempts to contextualize practitioners' experiences concerning smart meters use within mini-grid operations in rural areas using narrative analysis.

Against the above backdrop, the findings from this study offer new insights into smartmetering technologies and decision-making processes concerning mini-grid operations in rural areas. For mini-grid operators and policymakers, the qualitative interviews provide useful insights about unforeseen operational challenges associated with ground-level smart meter implementation. In conjunction with the narrative analysis, the SMAA results illustrate that decision-making surrounding smart metering ought to consider context-specific conditions. Moreover, to ensure efficient utilization of scarce resources, the experiences of the social enterprise indicate that mini-grid operators need to carefully evaluate smart meters in terms of multiple criteria. Herein, the hierarchy of criteria presented in this study could help mini-grid operators to comprehend the complexity associated with smart meter implementation in the purview of universal electrification of rural areas, especially the last mile. Finally, the current study fills the research gap by providing a systematic and theoretical analysis of smart meter selection in the energy planning literature for pre-feasibility, considering their capital-intensive nature in mini-grid operations.

This study was limited as it focused on a single company case study. While this allowed for a detailed analysis of one company, it weakened the general applicability of the obtained results. Therefore, future work should include a broader study that looks at smart meter usage on a practical level, considering and comparing different mini-grid companies.

Author Contributions: Conceptualization, V.B.; Methodology, V.B., T.S., J.V.P. and R.L.; Software, R.L.; Formal analysis, T.S.; Writing—original draft, V.B.; Writing—review & editing, T.S., J.V.P. and R.L.; Visualization, V.B.; Project administration, T.S., J.V.P. and R.L. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: The data is not accessible to the public at the request of the minigrid operator.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Energy Sector Management Assistance Program. Energy Analytics for Development: Big Data for Energy Access, Energy Efficiency, and Renewable Energy; World Bank: Washington, DC, USA, 2017.
- 2. Scheyvens, R.; Banks, G.; Hughes, E. The Private Sector and the SDGs: The Need to Move Beyond 'Business as Usual'. *Sustain*. *Dev.* **2016**, *24*, 371–382. [CrossRef]
- 3. Alam, M.; Bhattacharyya, S. Decentralized Renewable Hybrid Mini-Grids for Sustainable Electrification of the Off-Grid Coastal Areas of Bangladesh. *Energies* **2016**, *9*, 268. [CrossRef]
- 4. Bandi, V.; Sahrakorpi, T.; Paatero, J.; Lahdelma, R. The paradox of mini-grid business models: A conflict between business viability and customer affordability in rural India. *Energy Res. Soc. Sci.* **2022**, *89*, 102535. [CrossRef]
- 5. Cabanero, A.; Nolting, L.; Praktiknjo, A. Mini-Grids for the Sustainable Electrification of Rural Areas in Sub-Saharan Africa: Assessing the Potential of KeyMaker Models. *Energies* **2020**, *13*, 6350. [CrossRef]
- 6. Williams, N.J.; Jaramillo, P.; Taneja, J.; Ustun, T.S. Enabling private sector investment in microgrid-based rural electrification in developing countries: A review. *Renew. Sustain. Energy Rev.* **2015**, *52*, 1268–1281. [CrossRef]
- Batchelor, S.; Brown, E. Editorial for Special Issue: "Clean Energy Innovations: Challenges and Strategies for Low and Middle Income Countries". *Energies* 2021, 14, 8577. [CrossRef]
- 8. Proietti, S.; Sdringola, P.; Castellani, F.; Astolfi, D.; Vuillermoz, E. On the contribution of renewable energies for feeding a high altitude Smart Mini Grid. *Appl. Energy* **2017**, *185*, 1694–1701. [CrossRef]
- 9. Bayindir, R.; Colak, I.; Fulli, G.; Demirtas, K. Smart grid technologies and applications. *Renew. Sustain. Energy Rev.* 2016, 66, 499–516. [CrossRef]
- 10. Barai, G.R.; Krishnan, S.; Venkatesh, B. Smart metering and functionalities of smart meters in smart grid—A review. In Proceedings of the 2015 IEEE Electrical Power and Energy Conference (EPEC), London, ON, Canada, 26–28 October 2015; pp. 138–145.
- 11. Mugyenyi, J.; Mugume, E.; Williams, N.J.; Kimani, J.; Campbell, K.; Uche, E.; Dougherty, J. Smart metering technologies for mini grids in Africa: An Overview. In Proceedings of the 2021 IEEE PES/IAS PowerAfrica, Virtual, 23–27 August 2021; pp. 1–5.
- 12. Office of Electricity Delivery and Energy Reliability. *Advanced Metering Infrastructure and Customer Systems*; U.S. Department of Energy: Washington, DC, USA, 2016.
- Rahman, M.d.M.; Paatero, J.V.; Lahdelma, R. Evaluation of choices for sustainable rural electrification in developing countries: A multicriteria approach. *Energy Policy* 2013, 59, 589–599. [CrossRef]
- 14. Rojas-Zerpa, J.C.; Yusta, J.M. Methodologies, technologies and applications for electric supply planning in rural remote areas. *Energy Sustain. Dev.* **2014**, 20, 66–76. [CrossRef]
- 15. Lahdelma, R.; Salminen, P.; Hokkanen, J. Using Multicriteria Methods in Environmental Planning and Management. *Environ. Manag.* **2000**, *26*, 595–605. [CrossRef]
- 16. Løken, E. Use of multicriteria decision analysis methods for energy planning problems. *Renew. Sustain. Energy Rev.* 2007, 11, 1584–1595. [CrossRef]
- 17. Ali, T.; Nahian, A.J.; Ma, H. A hybrid multi-criteria decision-making approach to solve renewable energy technology selection problem for Rohingya refugees in Bangladesh. *J. Clean. Prod.* **2020**, *273*, 122967. [CrossRef]
- Micangeli, A.; Del Citto, R.; Kiva, I.N.; Santori, S.G.; Gambino, V.; Kiplagat, J.; Viganò, D.; Fioriti, D.; Poli, D. Energy Production Analysis and Optimization of Mini-Grid in Remote Areas: The Case Study of Habaswein, Kenya. *Energies* 2017, 10, 2041. [CrossRef]
- Nsafon, B.E.K.; Butu, H.M.; Owolabi, A.B.; Roh, J.W.; Suh, D.; Huh, J.-S. Integrating multi-criteria analysis with PDCA cycle for sustainable energy planning in Africa: Application to hybrid mini-grid system in Cameroon. *Sustain. Energy Technol. Assess.* 2020, 37, 100628. [CrossRef]
- 20. Rojas-Zerpa, J.C.; Yusta, J.M. Application of multicriteria decision methods for electric supply planning in rural and remote areas. *Renew. Sustain. Energy Rev.* 2015, *52*, 557–571. [CrossRef]
- 21. Peters, J.; Sievert, M.; Toman, M.A. Rural electrification through mini-grids: Challenges ahead. *Energy Policy* **2019**, *132*, 27–31. [CrossRef]
- 22. Lahdelma, R.; Hokkanen, J.; Salminen, P. SMAA-Stochastic multiobjective acceptability analysis. *Eur. J. Oper. Res.* **1998**, *106*, 137–143. [CrossRef]
- 23. Linkov, I.; Moberg, E.; Trump, B.D.; Yatsalo, B.; Keisler, J.M. *Multi-Criteria Decision Analysis: Case Studies in Engineering and the Environment*, 2nd ed.; CRC Press: Boca Raton, FL, USA, 2020; ISBN 978-0-429-32644-8.
- 24. Boje, D.M. *Narrative Methods for Organizational and Communication Research;* Sage Series in Management Research; Sage: Newcastle upon Tyne, UK, 2001; ISBN 978-0-7619-6586-2.
- Mills, A.; Durepos, G.; Wiebe, E. Encyclopedia of Case Study Research; SAGE Publications, Inc.: Thousand Oaks, CA, USA, 2010; ISBN 978-1-4129-5670-3.
- 26. Moezzi, M.; Janda, K.B.; Rotmann, S. Using stories, narratives, and storytelling in energy and climate change research. *Energy Res. Soc. Sci.* 2017, *31*, 1–10. [CrossRef]
- 27. Gill-Wiehl, A.; Miles, S.; Wu, J.; Kammen, D.M. Beyond customer acquisition: A comprehensive review of community participation in mini grid projects. *Renew. Sustain. Energy Rev.* 2022, 153, 111778. [CrossRef]
- 28. IRENA. *Quality Infrastructure for Smart Mini-Grids;* International Renewable Energy Agency: Abu Dhabi, United Arab Emirates, 2020; ISBN 978-92-9260-278-9.

- Nerini, F.F.; Broad, O.; Mentis, D.; Welsch, M.; Bazilian, M.; Howells, M. A cost comparison of technology approaches for improving access to electricity services. *Energy* 2016, 95, 255–265. [CrossRef]
- Mandelli, S.; Barbieri, J.; Mereu, R.; Colombo, E. Off-grid systems for rural electrification in developing countries: Definitions, classification and a comprehensive literature review. *Renew. Sustain. Energy Rev.* 2016, 58, 1621–1646. [CrossRef]
- 31. Alam, M.; Bhattacharyya, S. Are the off-grid customers ready to pay for electricity from the decentralized renewable hybrid mini-grids? A study of willingness to pay in rural Bangladesh. *Energy* **2017**, *139*, 433–446. [CrossRef]
- 32. Farhangi, H. The path of the smart grid. *IEEE Power Energy Mag.* **2010**, *8*, 18–28. [CrossRef]
- Tuballa, M.L.; Abundo, M.L. A review of the development of Smart Grid technologies. *Renew. Sustain. Energy Rev.* 2016, 59, 710–725. [CrossRef]
- 34. Welsch, M.; Bazilian, M.; Howells, M.; Divan, D.; Elzinga, D.; Strbac, G.; Jones, L.; Keane, A.; Gielen, D.; Balijepalli, V.S.K.M.; et al. Smart and Just Grids for sub-Saharan Africa: Exploring options. *Renew. Sustain. Energy Rev.* **2013**, *20*, 336–352. [CrossRef]
- 35. Khan, F. Scaling innovation. Nat. Energy 2018, 3, 910–912. [CrossRef]
- 36. Faruqui, A.; Sergici, S.; Sharif, A. The impact of informational feedback on energy consumption—A survey of the experimental evidence. *Energy* **2010**, *35*, 1598–1608. [CrossRef]
- Depuru, S.S.S.R.; Wang, L.; Devabhaktuni, V. Electricity theft: Overview, issues, prevention and a smart meter based approach to control theft. *Energy Policy* 2011, 39, 1007–1015. [CrossRef]
- Wagemann, B.; Manetsgruber, D. Risk Management for Mini-grid Deployment in Rural Areas. *Energy Procedia* 2016, 103, 106–110. [CrossRef]
- 39. Kirchhoff, H.; Strunz, K. Key drivers for successful development of peer-to-peer microgrids for swarm electrification. *Appl. Energy* **2019**, 244, 46–62. [CrossRef]
- 40. Shakya, B.; Bruce, A.; MacGill, I. Survey based characterisation of energy services for improved design and operation of standalone microgrids. *Renew. Sustain. Energy Rev.* 2019, 101, 493–503. [CrossRef]
- 41. Rashed Mohassel, R.; Fung, A.; Mohammadi, F.; Raahemifar, K. A survey on Advanced Metering Infrastructure. *Int. J. Electr. Power Energy Syst.* **2014**, *63*, 473–484. [CrossRef]
- 42. Andreadou, N.; Guardiola, M.O.; Fulli, G. Telecommunication Technologies for Smart Grid Projects with Focus on Smart Metering Applications. *Energies* 2016, *9*, 375. [CrossRef]
- 43. Numminen, S.; Lund, P.D. Evaluation of the reliability of solar micro-grids in emerging markets—Issues and solutions. *Energy Sustain. Dev.* **2019**, *48*, 34–42. [CrossRef]
- 44. Louie, H. Other Considerations. In *Off-Grid Electrical Systems in Developing Countries*; Louie, H., Ed.; Springer International Publishing: Cham, Switzerland, 2018; pp. 461–467. ISBN 978-3-319-91890-7.
- 45. ESMAP. Mini Grids for Half a Billion People: Market Outlook and Handbook for Decision Makers; World Bank: Washington, DC, USA, 2022.
- 46. Bhattacharyya, S.C. Mini-Grids for the Base of the Pyramid Market: A Critical Review. Energies 2018, 11, 813. [CrossRef]
- 47. Wątróbski, J.; Jankowski, J.; Ziemba, P.; Karczmarczyk, A.; Zioło, M. Generalised framework for multi-criteria method selection. *Omega* **2019**, *86*, 107–124. [CrossRef]
- Pesola, A.; Serkkola, A.; Lahdelma, R.; Salminen, P. Multicriteria evaluation of alternatives for remote monitoring systems of municipal buildings. *Energy Build.* 2014, 72, 229–237. [CrossRef]
- 49. Linkov, I.; Moberg, E. *Multi-Criteria Decision Analysis: Environmental Applications and Case Studies*, 1st ed.; CRC Press: Boca Raton, FL, USA, 2011; ISBN 978-1-4398-5318-4. [CrossRef]
- 50. Press Information Bureau. Hundred Percent Household Electrification Achieved in 25 States. Available online: https://pib.gov. in/newsite/PrintRelease.aspx?relid=186988 (accessed on 3 March 2023).
- 51. IEA. India Energy Outlook 2021; World Energy Outlook Special Report; International Energy Agency: Paris, France, 2021.
- 52. Kennedy, R.; Mahajan, A.; Urpelainen, J. Crowdsourcing data on the reliability of electricity service: Evidence from a telephone survey in Uttar Pradesh, India. *Energy Policy* **2020**, *145*, 111746. [CrossRef]
- 53. Bijlani, D.; Aklin, M.; Blankenship, B.; Nandan, V.; Urpelainen, J. Jharkhand Rural Energy Access: Enduring Challenges in Quality, Affordability, and Billing; ISEP and Oak Foundation: Washington, DC, USA, 2002; p. 20036.
- 54. Goyal, S.; Sergi, B.S.; Kapoor, A. Emerging role of for-profit social enterprises at the base of the pyramid: The case of Selco. *J. Manag. Dev.* **2017**, *36*, 97–108. [CrossRef]
- Knuckles, J. Business models for mini-grid electricity in base of the pyramid markets. *Energy Sustain. Dev.* 2016, 31, 67–82.
 [CrossRef]
- 56. Bandi, V.; Sahrakorpi, T.; Paatero, J.; Lahdelma, R. Touching the invisible: Exploring the nexus of energy access, entrepreneurship, and solar homes systems in India. *Energy Res. Soc. Sci.* 2020, *69*, 101767. [CrossRef]
- 57. Steiner, A.; Farmer, J.; Bosworth, G. Rural social enterprise—Evidence to date, and a research agenda. *J. Rural Stud.* **2019**, *70*, 139–143. [CrossRef]
- 58. Fereday, J.; Muir-Cochrane, E. Demonstrating Rigor Using Thematic Analysis: A Hybrid Approach of Inductive and Deductive Coding and Theme Development. *Int. J. Qual. Methods* **2006**, *5*, 80–92. [CrossRef]
- Lahdelma, R.; Salminen, P. SMAA-2: Stochastic Multicriteria Acceptability Analysis for Group Decision Making. Oper. Res. 2001, 49, 444–454. [CrossRef]

- 60. Pelissari, R.; Oliveira, M.C.; Amor, S.B.; Kandakoglu, A.; Helleno, A.L. SMAA methods and their applications: A literature review and future research directions. *Ann. Oper. Res.* **2020**, *293*, 433–493. [CrossRef]
- 61. Tervonen, T.; Lahdelma, R. Implementing stochastic multicriteria acceptability analysis. *Eur. J. Oper. Res.* 2007, 178, 500–513. [CrossRef]
- Lahdelma, R.; Salminen, P. SMAA in Robustness Analysis. In *Robustness Analysis in Decision Aiding, Optimization, and Analytics*; Doumpos, M., Zopounidis, C., Grigoroudis, E., Eds.; International Series in Operations Research & Management Science; Springer International Publishing: Cham, Switzerland, 2016; pp. 1–20. ISBN 978-3-319-33121-8.
- 63. Lahdelma, R.; Miettinen, K.; Salminen, P. Ordinal criteria in stochastic multicriteria acceptability analysis (SMAA). *Eur. J. Oper. Res.* 2003, 147, 117–127. [CrossRef]
- 64. Tervonen, T.; Figueira, J.R. A survey on stochastic multicriteria acceptability analysis methods. *J. Multi Criteria Decis. Anal.* **2008**, *15*, 1–14. [CrossRef]
- 65. Kontu, K.; Rinne, S.; Olkkonen, V.; Lahdelma, R.; Salminen, P. Multicriteria evaluation of heating choices for a new sustainable residential area. *Energy Build*. **2015**, *93*, 169–179. [CrossRef]
- Menou, A.; Lahdelma, R.; Salminen, P. Multicriteria Decision Aiding for Planning Renewable Power Production at Moroccan Airports. *Energies* 2022, 15, 5271. [CrossRef]
- 67. Keeney, R.L.; Raiffa, H. *Decisions with Multiple Objectives: Preferences and Value Trade-Offs*, 1st ed.; Cambridge University Press: Cambridge, UK, 1993; ISBN 978-0-521-43883-4.
- Geiger, D.; Antonacopoulou, E. Narratives and Organizational Dynamics: Exploring Blind Spots and Organizational Inertia. J. Appl. Behav. Sci. 2009, 45, 411–436. [CrossRef]
- 69. Riessman, C.K. Narrative Methods for the Human Sciences; Sage: Newcastle upon Tyne, UK, 2008; ISBN 978-0-7619-2997-0.
- 70. Flick, U. An Introduction to Qualitative Research, 3rd ed.; Sage: Newcastle upon Tyne, UK, 2006; ISBN 978-1-4129-1145-0.
- 71. Sovacool, B.K.; Axsen, J.; Sorrell, S. Promoting novelty, rigor, and style in energy social science: Towards codes of practice for appropriate methods and research design. *Energy Res. Soc. Sci.* **2018**, 45, 12–42. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.